\documentclass[a4paper]{amsart}

\pdfoutput=1

\usepackage[utf8]{inputenc}
\usepackage[T1]{fontenc}
\usepackage{lmodern}
%\usepackage[french,english]{babel}
\usepackage{amsthm, amssymb, amsmath, amsfonts, mathrsfs}
\usepackage[colorlinks=true, pdfstartview=FitV, linkcolor=blue, citecolor=blue, urlcolor=blue,pagebackref=false]{hyperref}


%\usepackage{a4wide}  %\usepackage{fullpage}
%\usepackage{amscd, graphicx, epsfig, psfrag, enumerate, dsfont}

%\usepackage{color}

%\usepackage[pagewise]{lineno}
%\linenumbers

\usepackage{microtype}

%\usepackage[notcite,notref,color]{showkeys}
%\definecolor{labelkey}{gray}{.8}
%\definecolor{refkey}{gray}{.8}

\newtheorem{thm}{Theorem}[section]
\newtheorem{prop}[thm]{Proposition}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{cor}[thm]{Corollary}
\theoremstyle{remark}
\newtheorem{rem}[thm]{Remark}
%\theoremstyle{definition}


\renewcommand{\le}{\leqslant}
\renewcommand{\ge}{\geqslant}
\renewcommand{\subset}{\subseteq}
\newcommand{\mcl}{\mathcal}
\newcommand{\msf}{\mathsf}
\newcommand{\B}{\mathbb{B}}
\newcommand{\e}{\mathbf{e}}
\newcommand{\E}{\mathbb{E}}
\newcommand{\cE}{\mathcal{E}}
\newcommand{\expec}[1]{\left\langle #1 \right\rangle}
\newcommand{\Gh}{G_{\mathsf{h}}}
\newcommand{\cGh}{\mathcal{G}_{\mathsf{h}}}
\newcommand{\Ah}{A_{\mathsf{h}}}
\newcommand{\ah}{a_{\mathsf{h}}}
\newcommand{\Ahi}{A_{\mathsf{h},i}}
\newcommand{\uh}{u_{\mathsf{h}}}
%\newcommand{\Ah}{A_\mathrm{hom}}
\renewcommand{\L}{\mathscr{L}}
\newcommand{\K}{\mathscr{K}_\xi}
\newcommand{\xm}{_{\xi,\mu}}
\newcommand{\N}{\mathbb{N}}
\newcommand{\na}{\nabla}
\newcommand{\Ll}{\left}
\newcommand{\Rr}{\right}
\newcommand{\lhs}{left-hand side}
\newcommand{\rhs}{right-hand side}
\newcommand{\1}{\mathbf{1}}
\newcommand{\R}{\mathbb{R}}
\newcommand{\C}{\mathbb{C}}
\newcommand{\Q}{\mathbb{Q}}
\newcommand{\Z}{\mathbb{Z}}
%\newcommand{\z}{^{(0)}}
\newcommand{\z}{^{\circ}}
\renewcommand{\P}{\mathbb{P}}
\newcommand{\Qx}{\mathsf{Q}^{(\xi)}}
\newcommand{\Qxe}{\mathsf{Q}^{(\xi,e)}}
\newcommand{\ov}{\overline}
\newcommand{\un}{\underline}
\newcommand{\td}{\tilde}
\newcommand{\eps}{\varepsilon}
\renewcommand{\d}{{\mathrm{d}}}
\newcommand{\mclN}{\mathcal{N}}
\newcommand{\var}{\mathbb{V}\mathrm{ar}}
\newcommand{\mfk}{\mathfrak}
\newcommand{\dr}{\partial}
\renewcommand{\o}{\omega}
\newcommand{\on}{[\un{\o}]_n}
\renewcommand{\O}{\Omega}
\newcommand{\tp}{^{(\td{p})}}
\newcommand{\T}{\mathbb{T}}
\renewcommand{\div}{\mathop{\rm div}}



\DeclareMathOperator*{\osc}{Osc}
\DeclareMathOperator{\arsinh}{arsinh}
\DeclareMathOperator{\ent}{\msf{Ent}}

\numberwithin{equation}{section}
\renewcommand{\theequation}{\thesection .\arabic{equation}}

\title[Correlation structure in stochastic homogenization]{Correlation structure of the corrector in stochastic homogenization}
\author{Jean-Christophe Mourrat, Felix Otto}

\address[Jean-Christophe Mourrat]{Ecole normale sup\'erieure de Lyon, CNRS, Lyon, France}

\address[Felix Otto]{Max Planck Institute for Mathematics in the Sciences, Leipzig, Germany}


\begin{document}
\begin{thebibliography}{99}


\bibitem[AS14]{armsma} S.N.\ Armstrong, C.K.\ Smart. Quantitative stochastic homogenization of elliptic equations in nondivergence form. \emph{Arch.\ Ration.\ Mech.\ Anal.}\ \textbf{214}, 867--911 (2014).

\bibitem[Ba08]{bal1} G.\ Bal. Central limits and homogenization in random media. \emph{Multiscale Model.\ Simul.}\ \textbf{7} (2), 677--702 (2008).

\bibitem[Ba10]{bal2} G.\ Bal. Homogenization with large spatial random potential. \emph{Multiscale Model.\ Simul.}\ \textbf{8} (4), 1484--1510 (2010).

\bibitem[Ba11]{bal3} G.\ Bal. Convergence to homogenized or stochastic partial differential equations. \emph{Appl.\ Math.\ Res.\ Express} \textbf{2}, 215--241 (2011).

\bibitem[BGGJ12]{balgar2} G.\ Bal, J.\ Garnier, Y.\ Gu, W. Jing. Corrector theory for elliptic equations with long-range correlated random potential. \emph{Asymptot.\ Anal.}\ \textbf{77} (3-4), 123--145 (2012). 

\bibitem[BGMP08]{balgar} G.\ Bal, J.\ Garnier, S.\ Motsch, V.\ Perrier. Random integrals and correctors in homogenization. \emph{Asymptot.\ Anal.}\ \textbf{59} (1-2), 1--26 (2008).

\bibitem[BG13]{balgu-rev} G.\ Bal, Y.\ Gu. Limiting models for equations with large random potential; a review. Preprint, \url{http://www.columbia.edu/~yg2254/papers/Review-CMS.pdf} (2013).

\bibitem[BJ11]{baljin} G.\ Bal, W.\ Jing. Corrector theory for elliptic equations in random media with singular Green's function. Application to random boundaries. \emph{Commun.\ Math.\ Sci.}\ \textbf{9} (2), 383--411 (2011). 

\bibitem[BB07]{berbis} N.\ Berger, M.\ Biskup. Quenched invariance principle for simple random walk on percolation clusters. \emph{Probab.\ Theory Related Fields} \textbf{137} (1-2), 83-120 (2007). 

\bibitem[BS11]{bisspo} M.\ Biskup, H.\ Spohn. Scaling limit for a class of gradient fields with nonconvex potentials. \emph{Ann.\ Probab.}\ \textbf{39} (1), 224--251 (2011). 

\bibitem[Bo09]{boivin} D.\ Boivin. Tail estimates for homogenization theorems in random media. \emph{ESAIM Probab.\ Stat.}\ \textbf{13}, 51-69 (2009). 

\bibitem[BP04]{boupia} A.\ Bourgeat, A.\ Piatnitski. Approximations of effective coefficients in stochastic homogenization. \emph{Ann.\ Inst.\ H.\ Poincar\'e Probab.\ Statist.}\ \textbf{40} (2), 153-165 (2004). 

\bibitem[CS10]{cafsou} L.A.\ Caffarelli, P.E.\ Souganidis. Rates of convergence for the homogenization of fully nonlinear uniformly elliptic pde in random media. \emph{Invent.\ Math.}\ \textbf{180} (2), 301--360 (2010).

\bibitem[CI03]{capiof} P.\ Caputo, D.\ Ioffe. Finite volume approximation of the effective diffusion matrix: the case of independent bond disorder. \emph{Ann.\ Inst.\ H.\ Poincar\'e Probab.\ Statist.}\ \textbf{39} (3), 505-525 (2003).

\bibitem[CF15]{confah} J.\ Conlon, A.\ Fahim. Strong convergence to the homogenized limit of parabolic equations with random coefficients. \emph{Trans.\ Amer.\ Math.\ Soc.},\ to appear (2015).

%\bibitem[CN00a]{connadny} J.\ Conlon, A.\ Naddaf. Green's functions for elliptic and parabolic equations with random coefficients. \emph{New York J. Math.} \textbf{6}, 153--225 (2000).
%
%\bibitem[CN00b]{connadejp} J.\ Conlon, A.\ Naddaf. On homogenization of elliptic equations with random coefficients. \emph{Electron. J. Probab.} \textbf{5} (9), 58 pp (2000).

\bibitem[CS14]{conspe} J.\ Conlon, T.\ Spencer. Strong convergence to the homogenized limit of elliptic equations with random coefficients. \emph{Trans.\ Amer.\ Math.\ Soc.}\ \textbf{366}, 1257-1288 (2014). 
%
%\bibitem[De99]{del} T.\ Delmotte. Parabolic Harnack inequality and estimates of Markov chains on graphs. \emph{Rev. Mat. Iberoamericana} \textbf{15} (1), 181--232 (1999).
%
%\bibitem[DD05]{deldeu} T.\ Delmotte, J.-D.\ Deuschel. On estimating the derivatives of symmetric diffusions in stationary random environment, with applications to $\nabla \phi$ interface model. \emph{Probab. Theory Related Fields} \textbf{133} (3), 358--390 (2005). 

%
%\bibitem[EGMN12]{cemracs} A.-C.\ Egloffe, A.\ Gloria, J.-C.\ Mourrat, T.N.\ Nguyen. Random walk in random environment, corrector equation, and homogenized coefficients: from theory to numerics, back and forth. Preprint, arXiv:1211.1834.
%

\bibitem[FOP82]{fop} R.\ Figari, E.\ Orlandi, G.\ Papanicolaou. Mean field and Gaussian approximation for partial differential equations with random coefficients. \emph{SIAM J.\ Appl.\ Math.}\ \textbf{42}, 1069--1077 (1982).

\bibitem[Fu05]{fun} T.\ Funaki. Stochastic interface models. \emph{Lectures on probability theory and statistics} 103--274, Ecole d'été de probabilités de Saint-Flour XXXIII, Lecture notes in mathematics \textbf{1869} (2005).

\bibitem[GOS01]{gos} G.\ Giacomin, S.\ Olla, H.\ Spohn. Equilibrium fluctuations for $\na \varphi$ interface model. \emph{Ann.\ Probab.}\ \textbf{29} (3), 1138--1172 (2001). 

\bibitem[GM12]{glmo} A.\ Gloria, J.-C.\ Mourrat. Spectral measure and approximation of homogenized coefficients. \emph{Probab.\ Theory Related Fields} \textbf{154} (1-2), 287--326 (2012).

\bibitem[GNO15]{gno} A.\ Gloria, S.\ Neukamm, F.\ Otto. Quantification of ergodicity in stochastic homogenization: optimal bounds via spectral gap on Glauber dynamics. \emph{Invent.\ math.},\ to appear (2015).

\bibitem[GNO14]{gno-2scale} A.\ Gloria, S.\ Neukamm, F.\ Otto. An optimal quantitative two-scale expansion in stochastic homogenization of discrete elliptic equations. \emph{ESAIM Math.\ Model.\ Numer.\ Anal.}\ \textbf{48} (2), 325--346 (2014).

\bibitem[GO11]{glotto} A.\ Gloria, F.\ Otto. An optimal variance estimate in stochastic homogenization of discrete elliptic equations. \emph{Ann.\ Probab.}\ \textbf{39} (3), 779--856 (2011). 

\bibitem[GO12]{glotto2} A.\ Gloria, F.\ Otto. An optimal error estimate in stochastic homogenization of discrete elliptic equations. \emph{Ann.\ Appl.\ Probab.}\ \textbf{22} (1), 1--28 (2012).
%

\bibitem[GB12]{gubal-1d} Y.\ Gu, G.\ Bal. Random homogenization and convergence to integrals with respect to the Rosenblatt process. \emph{J.\ Differential Equations} \textbf{253} (4), 1069--1087 (2012).

\bibitem[GB15]{gubal} Y.\ Gu, G.\ Bal. Fluctuations of Parabolic Equations with Large Random Potentials. \emph{Stoch.\ Partial Differ.\ Equ.\ Anal.\ Comput.},\ to appear (2015).


\bibitem[HS94]{helsjo} B.\ Helffer, J.\ Sjöstrand.
On the correlation for Kac-like models in the convex case. \emph{J.\ Statist.\ Phys.}\ \textbf{74} (1-2), 349--409 (1994).

%\bibitem[JKO]{jko} V.V.~Jikov, S.M.~Kozlov, O.A.~Oleinik. \emph{Homogenization of differential operators and integral functionals}. Translated from Russian by G.A.\ Yosifian. Springer (1994).
%
%\bibitem[Ka]{kato} T.\ Kato. \emph{Perturbation theory for linear operators} (2nd edition). Grundlehren der mathematischen Wissenschaften \textbf{132}, Springer (1976).
%
%\bibitem[KV86]{kipvar} C.\ Kipnis, S.R.S.\ Varadhan. Central limit theorem for additive functionals of reversible Markov processes and applications to simple exclusions, \emph{Comm. Math. Phys.} \textbf{104}, 1-19 (1986).
%


%\bibitem[Ko78]{kozlov1} S.M.~Kozlov. Averaging of random structures. \emph{Dokl. Akad. Nauk SSSR} \textbf{241} (5), 1016-1019 (1978). English transl.~: \emph{Soviet Math. Dokl.} \textbf{19} (4), 950-954  (1978).
%
\bibitem[Kü83]{kuen} R.\ Künnemann. The diffusion limit for reversible jump processes on $\Z^d$ with ergodic random bond conductivities. \emph{Comm. Math. Phys.} \textbf{90} (1), 27--68 (1983). 
%


\bibitem[MO13]{marott} D.\ Marahrens, F.\ Otto. Annealed estimates on the Green's function. Preprint, arXiv:1304.4408v1 (2013).


%\bibitem[Ma]{maxwell} J.C.~Maxwell. Medium in which small spheres are uniformly disseminated. \emph{A treatise on electricity and magnetism}, 3d ed., part II, chapter IX, article 314. Clarendon press (1891).
%

\bibitem[Mi11]{mill} J.\ Miller. Fluctuations for the Ginzburg-Landau $\nabla \phi$ interface model on a bounded domain. \emph{Comm.\ Math.\ Phys.}\ \textbf{308} (3), 591--639 (2011).

%\bibitem[Mo11]{berry} J.-C.\ Mourrat. A quantitative central limit theorem for the random walk among random conductances. Preprint, arXiv:1105.4485v1.
%
\bibitem[Mo11]{vardecay} J.-C.\ Mourrat. Variance decay for functionals of the environment viewed by the particle. \emph{Ann.\ Inst.\ H.\ Poincaré Probab.\ Statist.}\ \textbf{47} (1), 294--327 (2011).
%
%\bibitem[Mo12]{martingale_CLT} J.-C.\ Mourrat. On the rate of convergence in the martingale central limit theorem. \emph{Bernoulli}, to appear.
%

\bibitem[Mo14]{homog} J.-C.\ Mourrat. Kantorovich distance in the martingale CLT and quantitative homogenization of parabolic equations with random coefficients. \emph{Probab.\ Theory Related Fields} \textbf{160} (1--2), 279--314 (2014).

\bibitem[Mo15]{dl-diff} J.-C.\ Mourrat. First-order expansion of homogenized coefficients under Bernoulli perturbations. \emph{J.\ Math.\ Pures Appl.} \textbf{103}, 68--101 (2015).

\bibitem[MG15]{fluct2} J.-C.\ Mourrat, Y.\ Gu. Scaling limit of fluctuations in stochastic homogenization. In preparation.

\bibitem[MN15]{fluct} J.-C.\ Mourrat, J.\ Nolen. Scaling limit of the corrector in stochastic homogenization. In preparation.

\bibitem[NS97]{nadspe} A.\ Naddaf, T.\ Spencer.
On homogenization and scaling limit of some gradient perturbations of a massless free field. \emph{Comm.\ Math.\ Phys.}\ \textbf{183} (1), 55--84 (1997). 

\bibitem[NS98]{nadspe-unpub} A.\ Naddaf, T.\ Spencer. Estimates on the variance of some homogenization problems. Unpublished manuscript (1998).

%\bibitem[Pa95]{pap} G.C.~Papanicolaou. Diffusion in random media. \emph{Surveys in applied mathematics} 205--255 (J.B.\ Keller et al.\ Eds), Plenum Press (1995).

\bibitem[PV81]{papvar} G.C.~Papanicolaou, S.R.S.~Varadhan. Boundary value problems with rapidly oscillating random coefficients. \emph{Random fields (Esztergom, 1979)} 835-873, Colloq. Math. Soc. J\'anos Bolyai \textbf{27}, North-Holland (1981). 
%

%
%\bibitem[SS04]{sid} V.\ Sidoravicius, A.-S.\ Sznitman. Quenched invariance principles for walks on clusters of percolation or among random conductances. \emph{Probab. Theory Related Fields} \textbf{129} (2), 219-244 (2004).
%

\bibitem[Sj96]{sjo} J.\ Sjöstrand. Correlation asymptotics and Witten Laplacians. \emph{Algebra i Analiz} \textbf{8} (1), 160--191 (1996). Engl.\ transl.\ \emph{St.\ Petersburg Math.\ J.}\ \textbf{8} (1), 123--147.

%\bibitem[St]{ray} J.W.~Strutt (3d Baron Rayleigh). On the influence of obstacles arranged in rectangular order upon the properties of a medium. \emph{Philos. mag.} \textbf{34}, 481-502 (1892).

%
%
%\bibitem[Yu80]{yuri1} V.V.~Yurinski\u{\i}. On a Dirichlet problem with random coefficients. \emph{Stochastic differential systems (Proc. IFIP-WG 7/1 Working Conf., Vilnius, 1978)} 344-353, Lecture Notes in Control and Information Sci. \textbf{25}, Springer (1980).
%
\bibitem[Yu86]{yuri} V.V.~Yurinski\u{\i}. Averaging of symmetric diffusion in a random medium. (Russian) \emph{Sibirsk. Mat. Zh.} \textbf{27} (4), 167-180 (1986). English transl.\ \emph{Siberian Math. J.} \textbf{27} (4), 603-613 (1986).



\end{thebibliography}

%

\begin{abstract}

Recently, the quantification of errors in the stochastic homogenization of divergence-form operators has witnessed important progress. Our aim now is to go beyond error bounds, and give precise descriptions of the effect of the randomness, in the large-scale limit. This paper is a first step in this direction. Our main result is to identify the correlation structure of the corrector, in dimension $3$ and higher. This correlation structure is similar to, but different from that of a Gaussian free field.

\bigskip

\noindent \textsc{MSC 2010:} 35B27, 35J15, 35R60, 82D30.

\medskip

\noindent \textsc{Keywords:} homogenization, random media, two-point correlation function.

\end{abstract}
%
\maketitle
%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Introduction}
\label{s:intro}

Consider the solution $u_\eps : \R^d \to \R$ of the equation 
$$
\Ll(1-\na \cdot A\Ll(\frac{\cdot}{\eps}\Rr) \na\Rr) u_\eps = f ,
$$
where $f$ is a bounded smooth function, $A$ is a random field of symmetric matrices on $\R^d$, and $\eps > 0$. If $A$ is uniformly elliptic and has a stationary ergodic law, then $u_\eps$ is known to converge as $\eps \to 0$ to $\uh$, the solution of
$$
\Ll(1-\na \cdot \Ah \na\Rr) \uh = f ,
$$
where $\Ah$ is the (constant in space, deterministic) \emph{homogenized matrix}. This asymptotic result becomes more interesting if we can
\begin{enumerate}
\item devise (provably) efficient techniques to compute the homogenized matrix;
\item estimate the error in the convergence of $u_\eps$ to $\uh$.
\end{enumerate}
Doing so requires to introduce some additional assumption on the type of correlations displayed by the random coefficients; we assume from now on that they have a finite range of dependence. These problems were discussed in several works \cite{yuri, capiof,boupia,boivin,vardecay,conspe,confah} (see also \cite{cafsou, armsma} for non-divergence form operators), but optimal error bounds were worked out only recently in \cite{glotto, glotto2,glmo,gno} for (1), and in \cite{homog,gno-2scale} for (2) (in the discrete-space setting).

\medskip

While controlling the size of the errors in homogenization is useful, it would be better (and it is our aim) to describe precisely what the errors look like when $\eps$ is small. As an analogy, if the convergence of $u_\eps$ to $\uh$ is a law of large numbers, then we are looking for a central limit theorem.

The present paper is a first step towards this goal. In a discrete-space setting, it was proved in \cite{glotto} that stationary correctors exist for $d \ge 3$ (recall that we assume that the random coefficients have a finite range of dependence). In this case, let us write $\phi_\xi$ for the (stationary) corrector in the direction $\xi$ [see \eqref{defcorr}]. Under a minor smoothness assumption on the random coefficients, we show that for large~$x$, the correlation
$
\expec{\phi_\xi(0) \ \phi_\xi(x)}
$
becomes very close to 
\begin{equation}
\label{defK}
\K(x) := \int \na \cGh(y) \cdot \Qx \, \na \cGh(y-x) \ \d y,
\end{equation}
where $\cGh$ is the Green function of the homogenized operator $-\na \cdot \Ah \na$, and $\Qx$ is a $d\times d$ symmetric matrix that can be expressed in terms of correctors, see \eqref{defQx}. 

\medskip

This result paves the way for the understanding of the full scaling limit of $\eps^{-\frac{d-2}{2}} \phi(\cdot/\eps)$, seen as a random distribution. Indeed, the main missing ingredient is now to show that for any bounded, smooth test function $f$, the properly rescaled random variable $\sum_{x \in \Z^d} \phi(x) f(\eps x)$ converges in law to a Gaussian. This will be done in \cite{fluct}. 

This result on the corrector suggests (via a formal two-scale expansion) a scaling limit for $\eps^{-d/2} \Ll(u_\eps(\cdot)-\expec{u_\eps(\cdot)}\Rr)$ as well. This will be addressed in \cite{fluct2}.

\medskip

\noindent \textbf{Related works.} We now give a brief overview of related works. These can be divided into three groups. 

First, the questions that we consider here in dimension $d \ge 3$ have been investigated in dimension $1$. One can benefit from this setting to gain a better understanding of the effect of long-range correlations of the coefficients  \cite{balgar,gubal-1d}.

Second, similar questions have been explored for the homogenization of operators other than those considered here. Typically, one considers a deterministic operator perturbed by the addition of a rapidly oscillating random potential \cite{fop, bal1,bal2,bal3,baljin,balgar2,gubal}. We refer to \cite{balgu-rev} for a review.

Third, there is a deep connection between the corrector studied in the present paper and so-called $\na \varphi$ interface models \cite{fun}. At a heuristic level, one can think of the corrector as the zero-temperature limit of such an interface model (with a bond-dependent potential). The scaling limit of the interface model with convex, homogeneous potential was shown to be the Gaussian free field \cite{nadspe,gos,mill}. In view of this, one may expect (as was suggested in \cite[Conjecture~5]{berbis}) the correlations of the corrector to be described by a Gaussian free field as well. However, our results show that such is not the case in general. One way to see this is to observe that the Fourier transform of $\K$ is
\begin{equation}
\label{e:fourier}
\frac{p \cdot \Qx p}{(p \cdot \Ah p)^2} \qquad (p \in \R^d),
\end{equation}
while it should be of the form 
\begin{equation}
\label{e:fourier-gff}
\frac{1}{p \cdot B p} \qquad (p \in \R^d)
\end{equation}
for some symmetric, positive definite matrix $B$, if the correlations were those of a Gaussian free field. By considering coefficients with small ellipticity ratio, one can produce examples where \eqref{e:fourier} cannot be reduced to \eqref{e:fourier-gff}.

The proof given in \cite{nadspe} that the interface model rescales to the Gaussian free field (and the proof of the dynamical version of this in \cite{gos}) uses a Helffer-Sjöstrand representation of the correlations. We will also use this representation here, but with an important difference. In the case of the interface model, the Helffer-Sjöstrand representation readily enables to express the correlations of the interface as the averaged Green function of some operator, and the crux is then to show that this operator can be homogenized. In our case, the representation has a less clear interpretation. But it has to be so, since otherwise this would lead to Gaussian-free-field correlations.

Recently, a very interesting and direct connection was put forward in \cite{bisspo} between certain interface models with homogeneous but possibly non-convex potentials and the corrector considered here. The authors obtained the scaling limit of interface models with such potentials and zero tilt. They point out that the understanding of models with non-zero tilt could be obtained from the understanding of the scaling limit of the corrector. We refer to \cite[Section~6]{bisspo} for more on this.

\medskip
\noindent \textbf{Organization of the paper.} The precise setting and results of this paper are laid down in the next section. The Helffer-Sjöstrand representation of correlations is introduced in Section~\ref{s:HJ}. Section~\ref{s:green} recalls several crucial estimates on the corrector and the Green function. The goal of Section~\ref{s:two-scale} is to justify, in a weak sense, the two-scale expansion of the gradient of the Green function. The proof of the main result is then completed in Section~\ref{s:proof}.



%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Precise setting and results}
\label{s:setting}

We consider the (non-oriented) graph $(\Z^d, \B)$ with $d \ge 3$, where $\B$ is the set of nearest-neighbor edges. Let $(\e_1,\ldots,\e_d)$ be the canonical basis of $\Z^d$. For every edge $e \in \B$, there exists a unique pair $(\un{e},i) \in \Z^d \times \{1,\ldots d\}$ such that $e$ links $\un{e}$ to $\un{e} + \e_i$. Given such a pair, we write $\ov{e} = \un{e} + \e_i$. We call $\un{e}$ the \emph{base point} of the edge~$e$. For $f : \Z^d \to \R$, we let $\na f : \B \to \R$ be the gradient of $f$, defined by
$$
\na f(e) = f(\ov{e}) - f(\un{e}).
$$
We write $\na^*$ for the formal adjoint of $\na$, that is, for $F : \B \to \R$, $\na^*F : \Z^d \to \R$ is defined via
$$
(\na^* F)(x) = \sum_{i = 1}^d F((x-\e_i,x)) - F((x,x+\e_i)).
$$
For such $F$, we define $AF(e) = a_e F(e)$, where $(a_e)$ are real numbers taking values in a compact subset of $(0,+\infty)$. The operator of interest is $\na^*  A \na$. 

While a standard assumption for our purpose would be that $(a_e)$ are independent and identically distributed, the technicalities of the proof will be reduced by assuming that they are also smooth in the following sense. We give ourselves a family $(\zeta_e)_{e \in \B}$ of independent standard Gaussian random variables (we write $\P$ for the law of this family on $\Omega = \R^{\B}$, and $\expec{\cdot}$ for the associated expectation). The coefficients $(a_e)_{e \in \B}$ are then defined by $a_e = a(\zeta_e)$, where $a : \R \to \R$ is a fixed twice differentiable function with bounded first and second derivatives (and taking values in a compact subset of $(0,+\infty)$). 
%For a function $h: \R^d \to \R$, we write $\un{\dr} h : \R^d \to \R^d$ to denote its (continuous-space) gradient.
 
Under these conditions, it is well-known that there exists a constant matrix $\Ah$ such that $\na^*  A \na$ homogenizes over large scales to the continuous operator $-\na \cdot \Ah \na$. 


Let $\xi$ be a fixed vector of $\R^d$. For $\mu > 0$, let $\phi\xm$ be the unique stationary solution of
\begin{equation}
\label{defcorrmu}
\mu \phi\xm + \na^*  A (\xi + \na \phi\xm) = 0.
\end{equation}
It is proved in \cite{glotto} that (recall that we assume $d \ge 3$) $\phi\xm$ converges in $L^2(\Omega)$ to the unique stationary solution $\phi_\xi$ of
\begin{equation}
\label{defcorr}
\na^*  A (\xi + \na \phi_\xi) = 0.
\end{equation}
The function $\phi_\xi$ is called the (stationary) \emph{corrector} in the direction $\xi$. We use $\phi_i$ as shorthand for $\phi_{\e_i}$. In equations such as \eqref{defcorr}, $\xi$ is to be understood as the function from $\B$ to $\R$ such that $\xi(e) = \xi \cdot (\ov{e}-\un{e})$.

Let $\dr_e$ denote the weak derivative with respect to the random variable $\zeta_e$, which we may call a \emph{vertical derivative}. The formal adjoint of $\dr_e$ is
$$
\dr^*_e = -\dr_e + \zeta_e.
$$
We write $\dr f = (\dr_e f)_{e \in \B}$. For $F = (F_e)_{e \in \B}$, we write $\dr^* F = \sum_e \dr^*_e F_e$, and we let
\begin{equation}
\label{defL}
\L = \partial^* \partial.
\end{equation}
We write $|x|$ for the $L^2$-norm of $x \in \Z^d$. In order to keep light notation, we let $|x|_* = |x|+ 2$ (so that for instance $\log|x|_*$ is bounded away from $0$). 

\medskip

Here is our main result.

\begin{thm}[structure of correlations]
\label{t:struct}
Recall that we assume $d \ge 3$. Let $\mcl{E}_0$ be the set of edges with base-point $0 \in \Z^d$, let $\cGh : \R^d \to \R$ be the Green function of the (continuous-space) homogenized operator $- \na \cdot \Ah \na$, let $\Qx = (\Qx_{jk})_{1\le j,k\le d}$ be the matrix defined by
\begin{multline}
\label{defQx}
\Qx_{jk} = \\
\sum_{e \in \mcl{E}_0} \expec{ (\e_j + \na \phi_j)(e) (\xi + \na \phi_\xi)(e) \ \dr_e a_e (\L + 1)^{-1} \dr_e a_e \ (\e_k + \na \phi_k)(e)  (\xi + \na \phi_\xi)(e)},
\end{multline}
and let $\K(x)$ be defined by \eqref{defK}. There exists a constant $C < \infty$ such that for every $x \in \Z^d \setminus \{0\}$,
\begin{equation}
\label{e:struct}
\Ll| \expec{\phi_\xi(0) \ \phi_\xi(x)} - \K(x) \Rr| \le C \ \frac{\log^2 |x|_*}{|x|^{d-1}}.
\end{equation}
\end{thm}

\begin{rem}
When $\e_j$ is interpreted as a function over $\B$ as in \eqref{defQx}, it is to be understood as $\xi$ is in \eqref{defcorr}, that is, $\e_j(e)$ is $1$ if the edge $e$ is parallel to the basis vector $\e_j$, and is $0$ otherwise.
\end{rem}

\begin{rem}
The operator $\L$ is the infinitesimal generator of the Ornstein-Uhlenbeck semigroup on $\R^{\B}$, and $\P$ is a reversible measure for the associated dynamics. For more general distributions of coefficients, one may replace $\L$ by the infinitesimal generator of the Glauber dynamics, that is, to keep the definition \eqref{defL}, but with $\dr_e$ changed for
$$
\dr_e f= \E[f \, | \, (a_{e'})_{e' \neq e}] - f
$$
(in which case $(\L + 1)^{-1}$ must be replaced by $\L^{-1}$ in \eqref{defQx}). The setting we have chosen reduces the amount of technicality mostly by allowing us to use the chain rule for derivation.
\end{rem}

\begin{rem}
We learn from Proposition \ref{p:contract} and Theorem \ref{t:integrability} that the tensor $\Qx$ is well-defined.
%
From the identity
%
\begin{multline}
\label{F1}
\xi'\cdot\Qx\xi' = \\
\sum_{e \in \mcl{E}_0} \expec{(\xi' + \na \phi_{\xi'})(e)(\xi + \na \phi_\xi)(e) \ \dr_e a_e (\L + 1)^{-1} \dr_e a_e \
(\xi' + \na \phi_{\xi'})(e)(\xi + \na \phi_\xi)(e)},
\end{multline}
%
which follows from the linearity of $\phi_\xi$ in $\xi$,
we learn that $\Qx$ is positive semi-definite.
In particular, the Fourier transform of $\K$ is non-negative.

Moreover, $\Qx$ is non-degenerate as soon as the derivative of the function $a : \mathbb{R}\to\mathbb{R}$ is everywhere positive.
Indeed, if the expression (\ref{F1}) vanishes for $\xi'=\xi$, the strict positivity of the operator implies that for any $e\in\mcl{E}_0$,
$\dr_e a_e(\xi + \na \phi_\xi)^2(e)$ and thus $(\xi + \na \phi_\xi)^2(e)$ vanishes almost surely. This in turn implies that
%
$$
\sum_{e \in \mcl{E}_0} \expec{(\xi + \na \phi_\xi)(e)a_e(\xi + \na \phi_\xi)(e)}=\xi\cdot A_h\xi
$$
%
vanishes. By the non-degeneracy of the homogenized tensor $A_h$, this yields as desired $\xi=0$. The same argument also implies that
the null space of $\Qx$ is contained in the hyperplane orthogonal to $A_h\xi$. 
\end{rem}
\begin{rem}
There is no simple relation between the quartic form defined by $\Qx_{jk}$ and the quadratic form $A_h$,
besides that $\xi'\Qx\xi'$ is bounded from below by $(\xi'\cdot A_h\xi)^2$ up to a multiplicative constant.
As was noted in the introduction, $\K$ is not the Green function of a second-order operator in general. While its Fourier transform has
the right sign and homogeneity, it is not the inverse of a quadratic form.
\end{rem}
\begin{rem}
By polarization of the quartic form $\Qx_{jk}$ in the $\xi$-variables, 
one also obtains a result for covariances $\expec{\phi_\xi(0) \ \phi_{\xi'}(x)}$ with $\xi'\not=\xi$.
\end{rem}

%\begin{rem}
%\label{l:fourier}
%The Fourier transform (in the sense of distributions) of $\K$ is
%$$
%\frac{p \cdot \Qx p}{(p \cdot \Ah p)^2} \qquad (p \in \R^d).
%$$
%\end{rem}

\begin{rem}
We expect that at least if the environment is sufficiently mixing (as for instance when its correlations are of finite range), then there exists a matrix~$\Qx$ [whose explicit expression may differ from that given in \eqref{defQx}] such that the large-scale correlations of the correctors are described by \eqref{defK} and \eqref{e:struct}.
\end{rem}




%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Hellfer-Sjöstrand representation}
\label{s:HJ}
\begin{prop}[Helffer-Sjöstrand representation of correlations, \cite{helsjo,sjo,nadspe}]
Let $f,g : \Omega \to \R$ be centered square-integrable functions such that for every $e \in \B$, $\dr_e f$ and $\dr_e g$ are in $L^2(\Omega)$. We have
$$
\expec{f \ g} = \sum_{e \in \B} \expec{\dr_e f \ (\L+1)^{-1} \ \dr_e g}.
$$
\label{p:HJ}
\end{prop}
\begin{proof}
The claim is similar to (and simpler than) that obtained in \cite[Section~2.1]{nadspe}. We recall the proof briefly for the reader's convenience. By density, we can restrict our attention to functions $f, g$ that depend only on a finite number of $(\zeta_e)_{e \in \B}$, and also by density, we may assume $f$ and $g$ to be smooth functions.
Note that the commutator $[\dr_e, \dr_{e'}^*]$ satisfies
\begin{equation}
\label{commutator}
[\dr_e, \dr_{e'}^*] = \1_{e = e'}.
\end{equation}

Let us first assume that there exists a function $u \in L^2(\Omega)$ such that 
$g = \L u$. Writing $G = \dr u$, we observe that
\begin{eqnarray*}
\partial_e g & = & \partial_e \partial^* G \\
& = & \sum_{e'} \partial_e \partial_{e'}^* G_{e'} \\
& = & \sum_{e'} \Ll([\dr_e, \dr_{e'}^*] +  \partial_{e'}^* \partial_e \Rr) G_{e'} \\
& = & G_e + \sum_{e'} \partial_{e'}^* \partial_{e'} G_e,
\end{eqnarray*}
where we used \eqref{commutator} and the fact that $\partial_{e'} G_e = \partial_{e'} \partial_e u = \partial_e G_{e'}$ in the last step. Recalling the definition of $\L$ in \eqref{defL}, we arrive at
$$
\partial_e g = (\L + 1) G_e = (\L + 1) \partial_e u.
$$
In particular, $\partial_e u \in L^2(\Omega)$ and
$$
\expec{f \ g} = \expec{f \ \L u} = \sum_e \expec{\partial_e f \  \partial_e u} =\sum_e \expec{\dr_e f \ (\L+1)^{-1} \partial_e g}.
$$
In order to conclude, it suffices to check that %if $g \in L^2(\Omega)$ has zero average, then one can find a sequence $u_n \in L^2(\Omega)$ such that $g_n := \L u_n$ converges to $g$ in $L^2(\Omega)$.
the range of the operator $\L$ is dense in the set of centered square-integrable functions. If $g \in L^2(\Omega)$ is smooth, depends on a finite number of $(\zeta_e)_{e \in \B}$ and is in the orthogonal complement of $\textrm{Ran}(\L)$, then
$$
\expec{g \ \L g} = 0 = \sum_e \expec{|\dr_e g|^2},
$$
so $g$ is constant. It follows that the orthogonal complement of $\textrm{Ran}(\L)$ is the set of constant functions, and this completes the proof.
\end{proof}
The following additional information on $(\L+1)^{-1}$ will turn out to be useful.
\begin{prop}[contraction of $L^p$]
\label{p:contract}
For every $p \ge 2$, the operator $(\L+1)^{-1}$ is a contraction from $L^p(\Omega)$ to itself.
\end{prop}
\begin{proof}
Let $\Lambda$ be a finite subset of $\B$, and let $\mcl{F}_\Lambda$ the set of real functions of $(\zeta_e)_{e \in \Lambda}$. We define $H^1_\Lambda$ as the completion of the set of smooth functions in $\mcl{F}_\Lambda$ for the scalar product
$$
(u,v)_{H^1_\Lambda} = \expec{u \ v} + \sum_{e \in \Lambda} \expec{ \dr_e u \ \dr_e v}.
$$
For every $f \in H^1_\Lambda$, there exists a unique $u \in H^1_\Lambda$ such that
\begin{equation}
\label{weak}
\forall v \in H^1_\Lambda, \ (u,v)_{H^1_\Lambda} = \expec{f \ v},
\end{equation}
and this is nothing but the weak formulation of the equation $(\L+1)u = f$. For every $\eps > 0$, let
$
\psi_\eps(x) = \eps^{-1} \arctan(\eps x)
$
be a ``nice'' (in particular, bounded) approximation of the function $x \mapsto x$.
One can check that if $v \in H^1_\Lambda$, then $\psi_\eps(v \ |v|^{p-2}) \in H^1_\Lambda$. Hence, for $u \in H^1_\Lambda$ satisfying \eqref{weak},
$$
(u,\psi_\eps(u \ |u|^{p-2}))_{H^1_\Lambda} = \expec{f \ \psi_\eps(u \ |u|^{p-2})},
$$
and we recall that 
$$
(u,\psi_\eps(u \ |u|^{p-2}))_{H^1_\Lambda} = \expec{u \ \psi_\eps(u \ |u|^{p-2})} + \sum_{e \in \Lambda} \expec{ \dr_e u \ \dr_e \psi_\eps(u \ |u|^{p-2})}.
$$
Since $u \mapsto \psi_\eps(u \ |u|^{p-2})$ is an increasing function, it follows that for every $e$,  
$$
\expec{ \dr_e u \ \dr_e \psi_\eps(u \ |u|^{p-2})} \ge 0,
$$
and thus
$$
\expec{u \ \psi_\eps(u \ |u|^{p-2})} \le \expec{f \ \psi_\eps(u \ |u|^{p-2})}.
$$
By the monotone convergence theorem, the \lhs\ converges to $\expec{|u|^p}= \|u\|_p^p$ as $\eps$ tends to $0$. The \rhs\ is bounded by
$$
\|f\|_p \ \expec{|\psi_\eps(u \ |u|^{p-2})|^{p/{p-1}}}^{1-1/p}\le\|f\|_p\|u\|_p^{p-1},
$$
where we have used $|\psi_\eps(x)|\le|x|$. We have thus shown
$$
\|u\|_p^p \le \|f\|_p \ \|u\|_p^{p-1},
$$
that is, $\|u\|_p \le \|f\|_p$, and this implies the theorem.
\end{proof}
Using the fact that $(\L+1)^{-1}$ is a contraction on $L^2(\Omega)$, we deduce the following covariance estimate, which parallels those appearing in \cite{nadspe,nadspe-unpub} (Brascamp-Lieb inequality), \cite[Definition~1]{gno} and \cite[Lemma~3]{glotto2}.
\begin{cor}[covariance estimate]
\label{c:covar}
For $f$ and $g$ as in Proposition~\ref{p:HJ}, 
%$$
%\expec{f^2} \le \sum_{e \in \B} \expec{(\dr_e f)^2},
%$$
$$
\Ll| \expec{f \ g} \Rr| \le \sum_{e \in \B} \expec{(\dr_e f)^2}^{1/2} \expec{(\dr_e g)^2}^{1/2}.
$$
\end{cor}
%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Estimates on the corrector and the Green function}
\label{s:green}
The aim of this section is to gather several known estimates on the Green function and on the corrector. 

\begin{thm}[existence and integrability of the corrector \cite{glotto}]\label{t:integrability}
Recall that we assume $d \ge 3$. For every $\mu > 0$, there exists a unique stationary solution $\phi\xm$ to equation \eqref{defcorrmu}. 
Moreover, for every $p \ge 1$, $\expec{|\phi\xm(0)|^p}$ and $\expec{|\na \phi\xm(e)|^p}$ ($e \in \B$) are uniformly bounded in $\mu > 0$. The limit
\label{t:corrector}
$$
\phi_\xi = \lim_{\mu \to 0} \phi\xm
$$
is well-defined in $L^p(\Omega)$ and is the unique stationary solution to \eqref{defcorr}.
\end{thm}
A direct consequence of this result is:
\begin{cor}[almost-sure control of the corrector] Let $B_n = \{-n,\ldots,n\}^d$ and let $\B_n$ be the set of edges whose base-point is in $B_n$. For every $\beta > 0$, almost surely,
\label{c:corrector-growth}
$$
\lim_{n \to +\infty} \ n^{-\beta} \max_{x \in B_n} |\phi_\xi(x)| = 0
$$
and
$$
\lim_{n \to +\infty} \ n^{-\beta} \max_{e \in \B_n} |\na \phi_\xi(e)| = 0.
$$
\end{cor}
\begin{proof}
Let $p \ge 1$. By Chebyshev's inequality,
$$
\P[|\phi_\xi(0)| \ge x] \le \frac{\E[|\phi_\xi(0)|^p]}{x^p} \qquad (x > 0),
$$
so for any $\eps > 0$, by a union bound,
$$
\P\Ll[n^{-\beta} \max_{x \in B_n} |\phi_\xi(x)| \ge \eps\Rr] \le |B_n| \frac{\E[|\phi_\xi(0)|^p]}{(\eps n^\beta)^p}.
$$
The first part of the corollary follows by taking $p$ large enough and applying the Borel-Cantelli lemma. The second part is obtained in the same way.
\end{proof}

We write $G(x,y)$ for the Green function between points $x$ and $y$ in $\Z^d$, i.e.\ $G(x,y) = \Ll(\na^*  A \na\Rr)^{-1}(x,y)$ (the dependence on $(a_e)_{e \in \B}$ is kept implicit in the notation). For $\mu > 0$, we also let $G_\mu(x,y) = \Ll(\mu + \na^*  A \na\Rr)^{-1}(x,y)$.

Regularity theory ensures the following decay properties of the Green function (see for instance \cite[Proposition~3.6]{dl-diff} for a proof adapted to our context).
\begin{thm}[pointwise estimates on the Green function]
There exist $C<\infty$, $c > 0$ and $\alpha > 0$ such that for every $\mu \in [0,1/2]$ and $\zeta \in \Omega$,
\label{t:pointGreen}
\begin{equation}
\label{pointGreen}
G_\mu(0,x) \le \frac{C}{|x|_*^{d-2}} \ e^{-{c} \sqrt{\mu} |x|} \qquad (x \in \Z^d),
\end{equation}
\begin{equation}
\label{pointgradGreen}
\Ll|\na G_\mu(0,e)\Rr| \le \frac{C}{|\un{e}|_*^{d-2+\alpha}} \ e^{-{c} \sqrt{\mu} |\un{e}|} \qquad (e \in \B).
\end{equation}
\end{thm}
It was recently shown in \cite{marott} that, after averaging over the environment, the rates of decay of the gradient and mixed second gradient of the Green function behave as in the homogeneous case (see also \cite[Remark~11.2]{dl-diff} for the fact that the estimates hold uniformly over $\mu$).
\begin{thm}[annealed estimates on the gradients of the Green function \cite{marott}]
For every $1\le p <\infty$, there exists $C<\infty$ such that for every $\mu \in [0,1/2]$  and every $e,e' \in \B$,
$$
\expec{\Ll| \na G_\mu(0,e) \Rr|^p}^{1/p} \le \frac{C}{|\un{e}|_*^{d-1}},
$$
$$
\expec{\Ll| \na \na G_\mu(e,e') \Rr|^p}^{1/p} \le \frac{C}{|\un{e'}-\un{e}|_*^{d}}.
$$
\label{t:marott}
\end{thm}
\begin{rem}
Notice that $\na G(x,e)$ (for $x \in \Z^d$ and $e \in \B$) denotes the gradient of $G(x,\cdot)$ evaluated at the edge $e$. Similarly, $\na \na G(e,e')$ denotes the gradient of $\na G(\cdot,e')$ evaluated at the edge $e$.
\end{rem}
We conclude this section by recalling useful computations of vertical derivatives. The following two propositions are borrowed from \cite[Lemmas~2.4 and 2.5]{glotto}. 
\begin{prop}[derivatives of the corrector \cite{glotto}]
\label{p:deriv-corr}
For every $\mu > 0$, $x \in \Z^d$ and $e \in \B$, the approximate corrector $\phi\xm(x)$ is differentiable with respect to $\zeta_e$ and
$$
\dr_e \phi\xm(x) = - \dr_e a_e \na G_\mu(x,e) (\xi + \na \phi\xm)(e).
$$
\end{prop}
\begin{rem}
Recalling that we assume $a_e$ to be of the form $a(\zeta_e)$ with $a$ differentiable, we can rewrite $\dr_e a_e$ as $a'(\zeta_e)$.
\end{rem}
\begin{rem}
\label{r:phixmphimu}
Contrary to $\phi\xm$, the corrector $\phi_\xi$ is not well-defined for every value of $(\zeta_e) \in \Omega$, but only on a set of full probability measure. In order to prove a statement similar to Proposition~\ref{p:deriv-corr} for $\phi_\xi$ instead of $\phi\xm$, it is thus necessary to show first that $\phi_\xi$ is defined on a subset of $\Omega$ large enough that speaking of $\dr_e \phi_\xi$ be meaningful. We will however not show this here, since for our purpose, it is always possible to bypass this problem by approximating $\phi_\xi$ by $\phi\xm$, computing the derivatives, and then passing to the limit $\mu \to 0$. 
\end{rem}
\begin{prop}[derivatives of the Green function \cite{glotto}]
For every $\mu \ge 0$, $x,y \in \Z^d$ and $e \in \B$, the Green function $G_\mu(x,y)$ is differentiable with respect to $\zeta_e$ and
\label{p:deriv-green}
$$
\dr_e G_\mu(x,y) = - \dr_e a_e \na G_\mu(x,e) \na G_\mu(y,e).
$$
\end{prop}
These two propositions can be proved by differentiating the defining equation of, respectively, the corrector and the Green function, namely
$$
\mu \phi\xm + \na^* A (\xi + \na \phi\xm) = 0,
$$
$$
(\mu + \na^* A \na) G_\mu(x,\cdot) = \1_x.
$$
We refer to \cite{glotto} for details.

%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Two-scale expansion of the Green function}
\label{s:two-scale}
Note that since we assume the coefficients to be independent and identically distributed, the law of the coefficients is invariant under the rotations that preserve the lattice, and $\Ah$ is thus a multiple of the identity, say $\Ah = \ah \mathrm{Id}$. We define the discrete homogenized Green function $\Gh$ as the unique bounded solution of the equation
$$
\na^*  \Ah \na \Gh = \1_0,
$$
where $\Ah$ in the formula above acts as the multiplication by $\ah$ on every edge.
%For $f : \Z^d \to \R$ and $e \in \B$, we define $\na_j f(e)$ to be $\na f(e)$ when the edge $e$ is parallel to the basis vector $\e_j$, and to be zero otherwise. For the same $f$ but with $x \in \Z^d$, we may write $\na_j f(x)$ to denote $f(x+\e_j) - f(x)$ (in other words, $\na_j f(x)$ is the value of $\na_j f(e)$ for the edge $e$ with base point at $x$ and pointing in the direction of $\e_j$).
For $f : \Z^d \to \R$ and $x \in \Z^d$, we write $\na_j f(x)$ to denote $f(x+\e_j) - f(x)$. If instead we take $e \in \B$, we understand $\na_j f(e)$ to mean $\na_j f(\un{e})$, that is, the gradient of $f$ along the edge parallel to the vector $\e_j$ having the same base-point as $e$. 

The goal of this section is to prove the following quantitative two-scale expansion of the gradient of the Green function.
\begin{thm}[quantitative two-scale expansion of the Green function]
For every $p > 2$, there exists $C<\infty$ such that the following holds. If $g : \Omega \to \R$ is in $L^p(\Omega)$ and is differentiable with respect to $\zeta_b$ with $\dr_b g \in L^p(\Omega)$ for every $b \in \B$, then for every $e \in \B$, 
\label{t:two-scale}
\begin{multline}
\label{e:two-scale}
\Ll| \expec{g \ \na G(0,e)} - \sum_{j = 1}^d \na_j \Gh(e) \expec{g \ (\e_j + \na \phi_j)(e)} \Rr| \\
\le C\Ll( \|g\|_p \frac{\log|\un{e}|_*}{|\un{e}|_*^d} + \sum_{\substack{y \in \Z^d \\ b \in \B}} \Ll\| \dr_b g \Rr\|_p \frac{1}{|\un{e}-y|_*^{d-1} \ |\un{b}-y|_*^{d} \ |y|_*^d} \Rr).
\end{multline}
\end{thm}
\begin{rem}
Applying Theorem~\ref{t:two-scale} with $g = 1$, we obtain that
$$
\Ll| \expec{\na G(0,e)} - \na \Gh(e)\Rr| 
\le C \ \frac{\log|\un{e}|_*}{|\un{e}|_*^d}.
$$
\end{rem}
\begin{rem}
\label{r:translation}
By translation, under the assumptions of Theorem~\ref{t:two-scale}, we also have
\begin{multline*}
\Ll| \expec{g \ \na G(x,e)} - \sum_{j = 1}^d \na_j \Gh(e-x) \expec{g \ (\e_j + \na \phi_j)(e)} \Rr| \\
\le C\Ll( \|g\|_p \frac{\log|\un{e}-x|_*}{|\un{e}-x|_*^d} + \sum_{\substack{y \in \Z^d \\ b \in \B}} \Ll\| \dr_b g \Rr\|_p \frac{1}{|\un{e}-y|_*^{d-1} \ |\un{b}-y|_*^{d} \ |y-x|_*^d} \Rr),
\end{multline*}
where $e-x$ denotes the translation of the edge $e$ by the vector $-x$.
\end{rem}
We define $z : \Z^d \to \R$ by
\begin{equation}
\label{defz}
z(x) = G(0,x) - \Gh(x) - \sum_{j = 1}^d \phi_j(x) \ \na_j \Gh(x).
\end{equation}
\begin{prop}[equation for $z$ \cite{papvar,gno-2scale}]
Let $A_i(x)$ stand for $a_{x,x+\e_i}$. Write $\na^2 \Gh$ for the matrix with entries $\na^*_i \na_j \Gh$ ($1 \le i,j \le d$). Let $R$ be the matrix with entries $(R_{ij})$ satisfying
\label{p:z-eq}
$$
(R - \Ah)_{ij} =  - \Ll[A_i(\1_j^i + \na_i \phi_j)\Rr](\cdot - \e_i) \qquad (1\le i,j\le d),
$$
where $\1^i_{j} = \1_{i = j}$. For $e \in \B$ in the direction of $\e_i$, let 
$$
h(e) = - \Ll(A \sum_{j = 1}^d \phi_j(\cdot + \e_i) \na \na_j \Gh \Rr)(e),
$$
and denote the $\R^{d\times d}$-scalar product of two matrices $M$ and $N$ by $M:N$ (that is, the sum of all terms after entry-wise product). We have
\begin{equation}
\label{e:z-eq}
\na^* A \na z = R : \na^2 \Gh + \na^* h.
\end{equation}
\end{prop}
\begin{rem}
\label{r:centredR}
The crucial feature of the right-hand side of \eqref{e:z-eq} is that it involves only the second derivatives of $\Gh$ (this is precisely what one aims for when defining~$z$). Another aspect that will turn out to be important for our purpose is that $\expec{R(x)} = 0$. This follows from the fact (see e.g.\ \cite[(3.17)]{kuen}) that the $(i,j)$-th entry of the homogenized matrix $\Ah$ is equal to
$$
\expec{A_i(\1_j^i + \na_i \phi_j)} = \expec{\e_i \cdot A(\e_j + \na \phi_j)}.
$$
\end{rem}
\begin{proof}
We follow the line of argument given in the first step of the proof of \cite[Theorem~1]{gno-2scale} 
(itself inspired by the first proof of \cite[Theorem~3]{papvar}). For $f : \Z^d \to \R$, we write $\na^*_i f(x) = f(x-\e_i) - f(x)$. 
To begin with, we observe that the following discrete Leibniz rules hold, for $f,g : \Z^d \to \R$:
$$
\na_i(fg) = (\na_i f) g + f(\cdot + \e_i) \na_i g,
$$
$$
\na^*_i(fg) = (\na^*_i f) g + f(\cdot - \e_i) \na^*_i g.
$$

Recall that by definition,
$$
\na^* A \na G(0,\cdot) = \1_0 = \na^* \Ah \na \Gh,
$$
and thus,
$$
\na^* A \na (G(0,\cdot) - \Gh) = \na^*(\Ah - A) \na \Gh.
$$
%Using the chain rules, the latter can be rewritten as
%\begin{equation}
%\label{z-eq0}
%\sum_{i = 1}^d -(\na_i^*A_i) \na \Gh + (\Ahi-A_i)(\cdot - \e_i) \na^*_i \na_i \Gh,
%\end{equation}
%
Writing $\Ahi$ for the $i$-th diagonal coefficient of the (diagonal) matrix $\Ah$, we can express the \rhs\ above as
$$
\sum_{i = 1}^d \na^*_i(\Ahi - A_i) \na_i \Gh.
$$
We now need to compute
\begin{equation}
\label{z-eq1}
\na^* A \na\Ll( \phi_j\ \na_j \Gh \Rr).
\end{equation}
By the Leibniz rule,
$$
\na_i\Ll( \phi_j\ \na_j \Gh \Rr) = (\na_i\phi_j) \na_j \Gh  + \phi_j(\cdot + \e_i) \na_i \na_j \Gh.
$$
Hence, the term in \eqref{z-eq1} is equal to
$$
\sum_{i = 1}^d \na^*_i\Ll[ A_i\Ll(\na_i\phi_j \ \na_j \Gh  + \phi_j(\cdot + \e_i) \na_i \na_j \Gh \Rr) \Rr].
$$
We can thus rewrite $\na^* A \na z$ as
\begin{multline*}
\sum_{i = 1}^d \Ll\{\na^*_i(\Ahi - A_i) \na_i \Gh - \sum_{j = 1}^d \na^*_i\Ll[ A_i\Ll(\na_i\phi_j \na_j \Gh  + \phi_j(\cdot + \e_i) \na_i \na_j \Gh \Rr) \Rr] \Rr\} \\
= \sum_{i=1}^d \Ll\{\Ahi \na^*_i \na_i \Gh   - \sum_{j = 1}^d \na^*_i\Ll[ A_i\Ll((\1^i_{j} + \na_i\phi_j) \na_j \Gh  + \phi_j(\cdot + \e_i) \na_i \na_j \Gh \Rr) \Rr] \Rr\},
\end{multline*}
where we used the fact that $\Ah$ is constant. By the definition of the corrector, we have
$$
\sum_{i=1}^d \na^*_i A_i(\1^i_{j} + \na_i\phi_j) = \na^* A (\e_j + \na \phi_j) = 0,
$$
so by the Leibniz rule,
$$
\sum_{i,j=1}^d \na^*_i \Ll[A_i(\1^i_{j} + \na_i\phi_j) \na_j \Gh \Rr] = \sum_{i,j=1}^d \Ll[A_i(\1^i_{j} + \na_i\phi_j)\Rr](\cdot - \e_i) \  \na^*_i \na_j \Gh,
$$
and the conclusion follows.
\end{proof}
As a consequence, we get the following representation for $z$.
\begin{prop}[representation for $z$]
\label{p:rep-z}
For every $x \in \Z^d$,
\begin{equation}
\label{e:rep-z}
z(x) = \sum_{y \in \Z^d} G(x,y) \Ll( R : \na^2\Gh \Rr)(y) + \sum_{b \in \B} \na G(x,b) h(b).
\end{equation}
\end{prop}
\begin{proof}
Let $\td{z}(x)$ denote the \rhs\ of \eqref{e:rep-z}, which is well-defined by Corollary~\ref{c:corrector-growth}. Letting $\ov{z} = z - \td{z}$, one can check thanks to Proposition~\ref{p:z-eq} that $\na^* A \na \ov{z} = 0$. In particular,
$$
\sum_{x \in B_n} \ov{z}(x) \na^* A \na \ov{z}(x) = 0.
$$
This sum differs from
$$
\sum_{e \in \B_n} \na \ov{z}(e) \cdot A \na \ov{z}(e)
$$
by no more than a constant times
\begin{equation}
\label{e:repz1}
\sum_{e \in \B_{n+1} \setminus \B_{n}} \Ll(|\ov{z}(\un{e})| + |\ov{z}(\ov{e})|\Rr) |\na \ov{z}(e)|.
\end{equation}
This sum tends to $0$ as $n$ tends to infinity. To see this, we come back to the definitions of $z$ and $\td{z}$, given respectively in \eqref{defz} and in the \rhs\ of \eqref{e:rep-z}. Using Corollary~\ref{c:corrector-growth}, Theorem~\ref{t:pointGreen} and Proposition~\ref{p:sums} of the appendix, we obtain that for every $\beta > 0$, almost surely,
$$
|z(x)| = o\Ll(\frac{1}{|x|^{d-2-\beta}}\Rr)  \qquad (|x| \to \infty),
$$
$$
|\na {z}(e)| = o\Ll(\frac{1}{|\un{e}|^{d-2+\alpha-\beta}}\Rr)\qquad (|\un{e}| \to \infty)
$$
(where $\alpha$ comes from Theorem~\ref{t:pointGreen}), and the same relations hold for $z$ replaced by $\td{z}$, and thus also for $z$ replaced by $\ov{z}$. Since $d \ge 3$, we can take $\beta > 0$ sufficiently small to ensure that $2(d-2)+\alpha-2\beta > d-1$, and we obtain that the sum in \eqref{e:repz1} tends to $0$ as $n$ tends to infinity.

To sum up, we obtained that
$$
\lim_{n \to +\infty} \sum_{e \in \B_n} \na \ov{z}(e) \cdot A \na \ov{z}(e) = 0.
$$
Since $A$ is positive definite, we conclude that $\ov{z}$ is a constant. Now, both $z$ and $\td{z}$ tend to $0$ at infinity, so in fact $\ov{z} = 0$, and this concludes the proof.
\end{proof}
\begin{proof}[Proof of Theorem~\ref{t:two-scale}]
Let us first see that it suffices to show that 
\begin{equation}
\label{e:z-two}
\Ll| \expec{g \ \na z(e)} \Rr| 
\le C\Ll( \|g\|_p \frac{\log|\un{e}|_*}{|\un{e}|_*^d} + \sum_{\substack{y \in \Z^d \\ b \in \B}} \Ll\| \dr_b g \Rr\|_p \frac{1}{|\un{e}-y|_*^{d-1} \ |y-\un{b}|_*^{d} \ |y|_*^d} \Rr).
\end{equation}
Note that, by the Leibniz rule,
$$
\na_i z(x) = \na_i G(0,x) - \na_i \Gh(x) - \sum_{j = 1}^d \left[\na_i \phi_j(x) \ \na_j \Gh(x)  + \phi_j(x+\e_i) \ \na_i \na_j \Gh(x) \right].
$$
In order to prove that \eqref{e:z-two} implies \eqref{e:two-scale}, it is thus sufficient to show that
\begin{equation}
\label{z-3}
\Ll| \expec{g \ \phi_j(x+\e_i) \ \na_i \na_j \Gh(x)} \Rr| \le C \|g\|_p \frac{\log|x|_*}{|x|_*^d}.
\end{equation}
This is true since $|\na_i \na_j \Gh(x)| \lesssim |x|_*^{-d}$,
$$
\Ll| \expec{g \ \phi_j(x+\e_i)} \Rr| \le \|g\|_2 \ \|\phi_j\|_2,
$$
$\|\phi_j\|_2$ is finite by Theorem~\ref{t:corrector}, and we assume $p \ge 2$.

We now turn to the proof of \eqref{e:z-two}. From Proposition~\ref{p:rep-z}, we learn that 
$$
\na z(e) = \sum_{y \in \Z^d} \na G(e,y) (R : \na^2 \Gh)(y) + \sum_{b \in \B} \na \na G(e,b) h(b).
$$
We now proceed to show that each of the two terms
\begin{equation}
\label{part1}
\sum_{y \in \Z^d} \Ll| \expec{g \ \na G(e,y) (R : \na^2 \Gh)(y)} \Rr|,
\end{equation}
\begin{equation}
\label{part2}
\sum_{b \in \B} \Ll| \expec{g \ \na \na G(e,b) h(b)} \Rr|
\end{equation}
is bounded by the \rhs\ of \eqref{e:z-two}. 

\medskip

\noindent \emph{Step I.1.} We begin with \eqref{part1}, which is the more delicate. As noted in Remark~\ref{r:centredR}, the random variable $R$ is centered, so the expectation appearing within the absolute value in \eqref{part1} is in fact a correlation. We thus wish to apply Corollary~\ref{c:covar} and write
\begin{multline}
\label{part1step1}
\Ll| \expec{g \ \na G(e,y) (R : \na^2 \Gh)(y)} \Rr| \\
\le \sum_{b \in \B} \expec{[\dr_b (g \ \na G(e,y))]^2}^{1/2} \expec{[\dr_b (R : \na^2 \Gh)(y)]^2}^{1/2}.
\end{multline}
However, recalling that
$$
(R - \Ah)_{ij}(y) =  - \Ll[A_i(\1_j^i + \na_i \phi_j)\Rr](y - \e_i),
$$
we see that a slight difficulty appears because we have not given a meaning to $\dr_b \phi_j$. As was anticipated in Remark~\ref{r:phixmphimu}, this need not bother us. If we formally extend Proposition~\ref{p:deriv-corr} to the case $\mu = 0$, we arrive at the formal expression 
\begin{multline}
\label{formal1}
\dr_b (R_{ij}(y))   
  =  \dr_b a_b\big(- \1_{b = (y-\e_i,y)}(\1_j^i + \na_i \phi_j)(y-\e_i) \\
+ A_i(y-\e_i) \na \na G(y-\e_i,b) (\xi + \na \phi_j)(b)\big).
\end{multline}
The point now is that although we do not wish to discuss the sense of \eqref{formal1} as a derivative, we can take it as a \emph{definition} of the random variable $\dr_b (R_{ij}(y))$, and observe that \eqref{part1step1} holds. To see this, we approximate the \lhs\ of \eqref{part1step1} by introducing a small mass $\mu > 0$. We introduce
$$
(\Ah^\mu)_{ij} = \expec{A_i(\1_j^i + \na_i \phi_{j,\mu})}
$$
and $R^{\mu}$ by setting
$$
(R^{\mu} - \Ah^{\mu})_{ij}(y) =  - \Ll[A_i(\1_j^i + \na_i \phi_{j,\mu})\Rr](y - \e_i)
$$
(where of course $\phi_{j,\mu} = \phi_{\e_j,\mu}$). We can now write the \lhs\ of \eqref{part1step1} as the limit as $\mu$ tends to $0$ of
$$
\Ll| \expec{g \ \na G(e,y) (R^\mu : \na^2 \Gh)(y)} \Rr|.
$$
Applying Proposition~\ref{p:HJ} on this term is now legitimate, and by Proposition~\ref{p:deriv-corr},
\begin{multline}
\label{non-formal1}
\dr_b (R^\mu_{ij}(y))   
  = \dr_b a_b\big( - \1_{b = (y-\e_i,y)}(\1_j^i + \na_i \phi_{j,\mu})(y-\e_i) \\
+  A_i(y-\e_i) \na \na G_\mu(y-\e_i,b) (\xi + \na \phi_{j,\mu})(b) \big).
\end{multline}
By taking the limit $\mu \to 0$, it follows that \eqref{part1step1} holds with $\dr_b R$ defined by \eqref{formal1}.


\medskip

\noindent \emph{Step I.2.} 
%Let us write $\td{R}_{b,ij}(y)$ for the \rhs\ of \eqref{formal1}. 
By Hölder's inequality, it follows from Theorems~\ref{t:corrector} and \ref{t:marott} that
$$
\expec{(\dr_b R : \na^2\Gh)^2(y) }^{1/2} \lesssim \frac{1}{|\un{b}-y|_*^d \ |y|_*^d},
$$
where $\lesssim$ stands for $\le$ up to a multiplicative constant that only depends on $d$ and the Lipschitz constant of $a$.
On the other hand, using Proposition~\ref{p:deriv-green}, we see that
\begin{eqnarray*}
\dr_b (g \ \na G(e,y)) & = & (\dr_b g) \na G(e,y) + g \ \dr_b \na G(e,y) \\
& = & (\dr_b g) \na G(e,y) - g \ (\dr_b a_b) \na \na G(e,b) \na G(y,b).
\end{eqnarray*}
Using Hölder's inequality (in conjunction with the strict inequality $p>2$) and Theorem~\ref{t:marott}, we are led to
$$
\expec{[\dr_b (g \ \na G(e,y))]^2}^{1/2} \lesssim \frac{\| \dr_b g \|_p}{|\un{e}-y|_*^{d-1}} + \frac{\|g\|_p}{|\un{b}-\un{e}|_*^d \ |\un{b}-y|_*^{d-1}}.
$$
So we obtain from \eqref{part1step1} the inequality
\begin{multline}
\label{e:estim1}
\Ll| \expec{g \ \na G(e,y) (R : \na^2 \Gh)(y)} \Rr| \\
\lesssim \sum_{b \in \B} \Ll( \frac{\| \dr_b g \|_p}{|\un{e}-y|_*^{d-1}} + \frac{\|g\|_p}{|\un{b}-\un{e}|_*^d \ |\un{b}-y|_*^{d-1}} \Rr) \frac{1}{|\un{b}-y|_*^d \ |y|_*^d},
\end{multline}
and the term appearing in \eqref{part1} is bounded (up to a constant) by
$$
\sum_{\substack{y \in \Z^d \\ b \in \B}} \Ll( \frac{\| \dr_b g \|_p}{|\un{e}-y|_*^{d-1}} + \frac{\|g\|_p}{|\un{b}-\un{e}|_*^d \ |\un{b}-y|_*^{d-1}} \Rr) \frac{1}{|\un{b}-y|_*^d \ |y|_*^d}.
$$
To see that this is bounded by the \rhs\ of \eqref{e:z-two}, it suffices to observe that
$$
\sum_{y \in \Z^d} \frac{1}{|\un{b}-y|_*^{2d-1} \  |y|_*^d} \lesssim \frac{1}{|\un{b}|_*^d}
$$
and
\begin{equation}
\label{logcorr}
\sum_{b \in \B} \frac{1}{|\un{b}-\un{e}|_*^d \ |\un{b}|_*^d} \lesssim \frac{\log |\un{e}|_*}{|\un{e}|_*^d}.
\end{equation}
These two facts are proved in Proposition~\ref{p:sums} of the appendix.


\medskip

\noindent \emph{Step II.} We now turn to the analysis of \eqref{part2}. We note that
$$
\sum_{b \in \B} \Ll| \expec{g \ \na \na G(e,b) h(b)} \Rr| \le \sum_{b \in \B} \|g\|_2 \expec{\Ll( \na \na G(e,b) h(b) \Rr)^2}^{1/2}.
$$
Using the explicit form of $h$ given by Proposition~\ref{p:z-eq} together with Theorems~\ref{t:corrector} and~\ref{t:marott}, we arrive at
$$
\expec{\Ll( \na \na G(e,b) h(b) \Rr)^2}^{1/2} \lesssim \frac{1}{|\un{b}-\un{e}|_*^d \ |\un{b}|_*^d}.
$$
In view of \eqref{logcorr}, we have shown that the term in \eqref{part2} is bounded by a constant times
$$
\|g\|_2 \ \frac{\log |\un{e}|_*}{|\un{e}|_*^d},
$$
which is a better bound than needed.
\end{proof}

%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Proof of Theorem~\ref{t:struct}}
\label{s:proof}
\begin{proof}[Proof of Theorem~\ref{t:struct}]
Our starting point is the identity
\begin{equation}
\label{start}
\expec{\phi_\xi(0) \phi_\xi(x)} = \sum_{e \in \B} \expec{\dr_e \phi_\xi(0) \ (\L + 1)^{-1} \ \dr_e \phi_\xi(x)},
\end{equation}
with
\begin{equation}
\label{defdrphi}
\dr_e \phi_\xi(y) = - \dr_e a_e \na G(y,e) (\xi + \na \phi_\xi)(e) \qquad (y \in \Z^d).
\end{equation}
As in Step I.1 of the proof of Theorem~\ref{t:two-scale}, we do not mean to discuss the meaning of $\dr_e \phi_\xi(y)$ as a derivative of $\phi_\xi(y)$. Rather, it suffices for our purpose to observe that the identity in \eqref{start} holds with $\dr_e \phi_\xi(0)$ and $\dr_e \phi_\xi(x)$ \emph{defined} by \eqref{defdrphi}. This follows easily by approximating $\phi_\xi$ by $\phi\xm$, applying Propositions~\ref{p:HJ} and~\ref{p:deriv-corr}, and letting $\mu$ tend to $0$. 

Replacing $\dr_e \phi_\xi(0)$ and $\dr_e \phi_\xi(x)$ by their definitions, the summand in the \rhs\ of \eqref{start} becomes
\begin{equation}
\label{e:summand}
\expec{\dr_e a_e \na G(0,e) (\xi + \na \phi_\xi)(e) \ (\L + 1)^{-1} \ \dr_e a_e \na G(x,e) (\xi + \na \phi_\xi)(e)}.
\end{equation}
We see that two $\na G$ terms appear in this expectation. We will ``pull out of the expectation'' each of these $\na G$ terms using Theorem~\ref{t:two-scale}. These form the two first steps of the proof. The last step discusses how to replace $\na \Gh$ by its continuous-space counterpart $\na \cGh$.

\medskip

\noindent \emph{Step 1.} Defining
$$
g_e(x) = \dr_e a_e  (\xi + \na \phi_\xi)(e) \ (\L + 1)^{-1} \ \dr_e a_e \na G(x,e) (\xi + \na \phi_\xi)(e),
$$
we see that we can rewrite the term in \eqref{e:summand} as
$$
\expec{g_e(x) \ \na G(0,e)},
$$
and we wish to justify that
\begin{equation}
\label{e:step1todo}
\sum_{e \in \B} \Ll|\expec{g_e(x) \ \na G(0,e)} - \sum_{j = 1}^d \na_j \Gh(e) \expec{g_e(x) \ (\e_j + \na \phi_j)(e)} \Rr| \lesssim \frac{\log^2|x|_*}{|x|_*^{d-1}}.
\end{equation}
In order to apply Theorem~\ref{t:two-scale} for this purpose, we need to compute $\dr_b g_e(x)$ for every $b \in \B$. From the commutation relation in~\eqref{commutator}, it follows that
$$
\dr_b \ \L = (\L+1) \ \dr_b,
$$
and thus
$$
\dr_b \ (\L+1)^{-1} = (\L+2)^{-1} \ \dr_b.
$$
From this observation, we get that
\begin{equation}
\label{e:decompgex}
\dr_b g_e(x) = g^{(1)}_{b,e}(x) + g^{(2)}_{b,e}(x) + g^{(3)}_{b,e}(x) + g^{(4)}_{b,e}(x)
\end{equation}
with
$$
g^{(1)}_{b,e}(x) = - \dr_e a_e \dr_b a_b \na \na G(e,b) (\xi + \na \phi_\xi)(b) \ (\L + 1)^{-1} \ \dr_e a_e \na G(x,e) (\xi + \na \phi_\xi)(e),
$$
\begin{multline*}
g^{(2)}_{b,e}(x) = - \dr_e a_e  (\xi + \na \phi_\xi)(e)  \ (\L + 2)^{-1} \\ 
\dr_e a_e\dr_b a_b \na \na G(e,b)\Ll[  \na G(x,b)(\xi + \na \phi_\xi)(e) + \na G(x,e)  (\xi + \na \phi_\xi)(b) \Rr],
\end{multline*}
$$
g^{(3)}_{b,e}(x) = \1_{e = b} \ \dr_e^2 a_e (\xi + \na \phi_\xi)(e) \ (\L + 1)^{-1} \ \dr_e a_e \na G(x,e) (\xi + \na \phi_\xi)(e),
$$
and
$$
g^{(4)}_{b,e}(x) = \1_{e = b} \ \dr_e a_e (\xi + \na \phi_\xi)(e) \ (\L +2)^{-1} \ \dr_e^2 a_e \na G(x,e) (\xi + \na \phi_\xi)(e).
$$
As before, we do not wish to discuss the meaning of \eqref{e:decompgex} as a derivative, but rather use the fact that if $\dr_b g_e(x)$ is defined in this way, then by the usual approximation argument,
\begin{multline*}
\Ll|\expec{g_e(x) \ \na G(0,e)} - \sum_{j = 1}^d \na_j \Gh(e) \expec{g_e(x) \ (\e_j + \na \phi_j)(e)} \Rr| \\
\lesssim \|g_e(x)\|_p \frac{\log|\un{e}|_*}{|\un{e}|_*^d} + \sum_{\substack{y \in \Z^d \\ b \in \B}} \Ll\| \dr_b g_e(x) \Rr\|_p \frac{1}{|\un{e}-y|_*^{d-1} \ |\un{b}-y|_*^{d} \ |y|_*^d}.
\end{multline*}
From Proposition~\ref{p:contract} and Theorems~\ref{t:corrector} and \ref{t:marott}, we learn that
$$
\|g_e(x)\|_p \lesssim \frac{1}{|\un{e}-x|_*^{d-1}}
$$
and
$$
\|\dr_b g_e(x)\|_p \lesssim \frac{1}{|\un{b} - \un{e}|_*^d} \Ll(\frac{1}{|\un{e}-x|_*^{d-1}} + \frac{1}{|\un{b}-x|_*^{d-1}}\Rr).
$$
Hence, up to a multiplicative constant, the \lhs\ of \eqref{e:step1todo} is smaller than the sum of the following two terms:
\begin{equation}
\label{e:term1}
\sum_{e \in \B} \frac{\log |\un{e}|_*}{|\un{e}-x|_*^{d-1} \ |\un{e}|_*^{d}},
\end{equation}
\begin{equation}
\label{e:term2}
\sum_{\substack{e,b \in \B \\ y \in \Z^d}}\frac{1}{|\un{b} - \un{e}|_*^d} \Ll(\frac{1}{|\un{e}-x|_*^{d-1}} + \frac{1}{|\un{b}-x|_*^{d-1}}\Rr) \frac{1}{|\un{e}-y|_*^{d-1} \ |\un{b}-y|_*^{d} \ |y|_*^d}.
\end{equation}
By Remark~\ref{r:logs} of the appendix, the sum in \eqref{e:term1} is dominated by a constant times the \rhs\ of \eqref{e:step1todo}. As for the sum in \eqref{e:term2}, we can further split it into the sum of
\begin{equation}
\label{e:term21}
\sum_{\substack{e,b \in \B \\ y \in \Z^d}}\frac{1}{|\un{b} - \un{e}|_*^d \ |\un{e}-x|_*^{d-1} \ |\un{e}-y|_*^{d-1} \ |\un{b}-y|_*^{d} \ |y|_*^d}
\end{equation}
and
\begin{equation}
\label{e:term22}
\sum_{\substack{e,b \in \B \\ y \in \Z^d}}\frac{1}{|\un{b} - \un{e}|_*^d \ |\un{b}-x|_*^{d-1} \ |\un{e}-y|_*^{d-1} \ |\un{b}-y|_*^{d} \ |y|_*^d}.
\end{equation}
By repeatedly applying Proposition~\ref{p:sums} of the appendix, we can bound the sum in \eqref{e:term21} by
$$
\sum_{\substack{e\in \B \\ y \in \Z^d}}\frac{\log|\un{e}-y|_*}{|\un{e}-y|_*^{2d-1} \ |\un{e}-x|_*^{d-1}  \ |y|_*^d} \lesssim \sum_{e \in \B} \frac{1}{|\un{e}-x|_*^{d-1}  \ |\un{e}|_*^d} \lesssim \frac{\log|x|_*}{|x|_*^{d-1}},
$$
and similarly, bound the sum in \eqref{e:term22} by
$$
\sum_{\substack{b \in \B \\ y \in \Z^d}}\frac{\log|\un{b}-y|_*}{|\un{b}-y|_*^{2d-1} \ |\un{b}-x|_*^{d-1}   \ |y|_*^d} \lesssim \sum_{b \in \B} \frac{1}{|\un{b}|_*^d \ |\un{b}-x|_*^{d-1} } \lesssim \frac{\log|x|_*}{|x|_*^{d-1}},
$$
and the proof of \eqref{e:step1todo} is complete.

\medskip

\noindent \emph{Step 2.} Recall that we have written $\expec{\phi_\xi(0) \ \phi_\xi(x)}$ as
$$
\sum_{e \in \B} \expec{g_e(x) \na G(0,e)},
$$
so we proved in Step 1 that
\begin{equation}
\label{e:step1result}
\Ll|\expec{\phi_\xi(0) \ \phi_\xi(x)} - \sum_{e \in \B} \sum_{j = 1}^d \na_j \Gh(e) \expec{g_e(x) \ (\e_j + \na \phi_j)(e)} \Rr| \lesssim \frac{\log^2|x|_*}{|x|_*^{d-1}}.
\end{equation}
We now aim to show that
\begin{multline}
\label{e:aim2}
\sum_{e \in \B} \sum_{j = 1}^d \Ll|\na_j \Gh(e) \expec{g_e(x) \ (\e_j + \na \phi_j)(e)} - \sum_{k = 1}^d \na_j \Gh(e) \Qxe_{jk} \na_k \Gh(e-x) \Rr| \\
\lesssim \frac{\log^2|x|_*}{|x|_*^{d-1}},
\end{multline}
where $\Qxe_{jk}$ is defined by
\begin{multline}
\label{defQxe}
\Qxe_{jk} = \\
\expec{\dr_e a_e  (\e_j + \na \phi_j)(e) (\xi + \na \phi_\xi)(e) \ (\L + 1)^{-1} \ \dr_e a_e (\e_k + \na \phi_k)(e)  (\xi + \na \phi_\xi)(e)}.
\end{multline}
For $j \in \{1,\ldots,d\}$, we let
$$
\td{g}_{e,j} = \dr_e a_e  (\xi + \na \phi_\xi)(e) \ (\L + 1)^{-1} \ \dr_e a_e (\e_j + \na \phi_j)(e)  (\xi + \na \phi_\xi)(e),
$$
and observe that since $(\L+1)^{-1}$ is symmetric,
\begin{equation}
\label{e:selfadj}
\expec{g_e(x) \ (\e_j + \na \phi_j)(e)} = \expec{ \td{g}_{e,j} \ \na G(x,e)}.
\end{equation}
We let
$$
\dr_b \td{g}_{e,j} = \td{g}^{(1)}_{b,e,j} + \td{g}^{(1)}_{b,e,j} + \td{g}^{(3)}_{b,e,j} + \td{g}^{(4)}_{b,e,j},
$$
where
$$
\td{g}^{(1)}_{b,e,j} = - \dr_e a_e \dr_b a_b \na \na G(e,b) (\xi + \na \phi_\xi)(b) \ (\L + 1)^{-1} \ \dr_e a_e (\e_j + \na \phi_j)(e)  (\xi + \na \phi_\xi)(e),
$$
\begin{multline*}
\td{g}^{(2)}_{b,e,j} = -\dr_e a_e \xi + \na \phi_\xi)(e) \ (\L + 2)^{-1} \\ \dr_e a_e\dr_b a_b\na \na G(e,b) \Ll[(\e_j + \na \phi_j)(b) (\xi + \na \phi_\xi)(e) + (\e_j + \na \phi_j)(e)(\xi + \na \phi_\xi)(b) \Rr],
\end{multline*}
$$
\td{g}^{(3)}_{b,e,j} = \1_{e=b} \ \dr_e^2 a_e (\xi + \na \phi_\xi)(e) \ (\L + 1)^{-1} \ \dr_e a_e (\e_j + \na \phi_j)(e)  (\xi + \na \phi_\xi)(e),
$$
and
$$
\td{g}^{(4)}_{b,e,j} = \1_{e=b} \ \dr_e a_e  (\xi + \na \phi_\xi)(e) \ (\L + 2)^{-1} \ \dr_e^2 a_e (\e_j + \na \phi_j)(e)  (\xi + \na \phi_\xi)(e).
$$
As before (and because of Remark~\ref{r:translation}), this definition ensures that
\begin{multline*}
\Ll| \expec{ \td{g}_{e,j} \ \na G(x,e)} - \sum_{k = 1}^d \na_k \Gh(e-x) \expec{\td{g}_{e,j} \ (\e_k + \na \phi_k)(e)} \Rr| \\
\lesssim \|\td{g}_{e,j}\|_p \frac{\log|\un{e}-x|_*}{|\un{e}-x|_*^d} + \sum_{\substack{y \in \Z^d \\ b \in \B}} \Ll\| \dr_b \td{g}_{e,j} \Rr\|_p \frac{1}{|\un{e}-y|_*^{d-1} \ |\un{b}-y|_*^{d} \ |y-x|_*^d}.
\end{multline*}
Moreover, we infer from Proposition~\ref{p:contract} and Theorems~\ref{t:corrector} and \ref{t:marott} that for any $1\le p<\infty$
(and thus in particular the $p>2$ needed above)
$$
\|\td{g}_{e,j}\|_p \lesssim 1
$$
and
$$
\Ll\| \dr_b \td{g}_{e,j} \Rr\|_p \lesssim \frac{1}{|\un{b}-\un{e}|_*^d}.
$$
Since
$$
\expec{\td{g}_{e,j} \ (\e_k + \na \phi_k)(e)} = \Qxe_{jk},
$$ 
we obtain that
\begin{multline*}
%\label{something}
\Ll| \expec{ \td{g}_{e,j} \ \na G(x,e)} - \sum_{k = 1}^d \Qxe_{jk} \na_k \Gh(e-x)  \Rr| \\
\lesssim \frac{\log|\un{e}-x|_*}{|\un{e}-x|_*^d} + \sum_{\substack{y \in \Z^d \\ b \in \B}} \frac{1}{|\un{b}-\un{e}|_*^d \ |\un{e}-y|_*^{d-1} \ |\un{b}-y|_*^{d} \ |y-x|_*^d},
\end{multline*}
and thus by \eqref{e:selfadj}, up to a multiplicative constant, the \lhs\ of \eqref{e:aim2} is smaller than
\begin{equation}
\label{e:rema}
\sum_{e \in \B} \frac{1}{|e|_*^{d-1}} \Ll(\frac{\log|\un{e}-x|_*}{|\un{e}-x|_*^d} +  \sum_{\substack{y \in \Z^d \\ b \in \B}} \frac{1}{|\un{b}-\un{e}|_*^d \ |\un{e}-y|_*^{d-1} \ |\un{b}-y|_*^{d} \ |y-x|_*^d} \Rr).
\end{equation}
From Remark~\ref{r:logs} of the appendix, we have
$$
\sum_{e \in \B} \frac{1}{|e|_*^{d-1}} \frac{\log|\un{e}-x|_*}{|\un{e}-x|_*^d} \lesssim \frac{\log^2|x|_*}{|x|_*^{d-1}}.
$$
The remaining sum from \eqref{e:rema} can be bounded, using Proposition~\ref{p:sums} repeatedly, by
$$
\sum_{\substack{y \in \Z^d \\ e \in \B}} \frac{\log |\un{e}-y|_*}{|\un{e}|_*^{d-1} \  |\un{e}-y|_*^{2d-1}  \ |y-x|_*^d}  \lesssim \sum_{y \in \Z^d} \frac{1}{|y|_*^{d-1} \ |y-x|_*^d} \lesssim \frac{\log|x|_*}{|x|_*^{d-1}},
$$
and this finishes the proof of \eqref{e:aim2}.

\medskip

\noindent \emph{Step 3.} Note that by the stationarity of the environment,  the matrix $\Qxe$ depends on the edge $e$ only through its orientation. On the other hand, the quantities $\na_j \Gh(e)$ and $\na_j \Gh(e-x)$ depend on the edge $e$ only through its base point. 
We also observe that the matrix $\Qx$ introduced in \eqref{defQx} is by definition $\sum_{e \in \mcl{E}_0} \mathsf{Q}^{(\xi,e)}$. Hence, the previous steps of the proof have led us, see \eqref{e:step1result} and \eqref{e:aim2}, to
\begin{equation}
\label{e:step2result}
\Ll|\expec{\phi_\xi(0) \ \phi_\xi(x)} - \sum_{y \in \Z^d} \sum_{j,k = 1}^d  \na_j \Gh(y) \Qx_{jk} \na_k \Gh(y-x)  \Rr| \lesssim \frac{\log^2|x|_*}{|x|_*^{d-1}}.
\end{equation}
In order to complete the proof of Theorem~\ref{t:struct}, it thus suffices to show that 
$$
\Ll| \sum_{y \in \Z^d} \sum_{j,k = 1}^d  \na_j \Gh(y) \Qx_{jk} \na_k \Gh(y-x) - \K(x) \Rr| \lesssim \frac{\log |x|_*}{|x|_*^{d-1}},
$$
where $\K$ was introduced in \eqref{defK}. We learn from Proposition~\ref{p:discrete-to-continuous} of the appendix that
$$
\Ll| \na_j \Gh(y)  - \frac{\dr \cGh}{\dr y_j}(y)  \Rr| \lesssim \frac{1}{|y|^d}.
$$
As a consequence, 
\begin{multline*}
 \sum_{\substack{y \in \Z^d \setminus \{0\} \\ 1 \le j,k \le d}}  \Ll| \na_j \Gh(y) \Qx_{jk} \na_k \Gh(y-x) -   \frac{\dr \cGh}{\dr y_j}(y)  \Qx_{jk} \na_k \Gh(y-x) \Rr| \\
\lesssim \sum_{y \in \Z^d \setminus \{0\}} \frac{1}{|y|^d \ |y-x|_*^{d-1}} \lesssim \frac{\log |x|_*}{|x|_*^{d-1}},
\end{multline*}
where we used Proposition~\ref{p:sums} of the appendix in the last step. Similarly,
$$
 \sum_{\substack{y \in \Z^d \setminus \{0,x\} \\ 1 \le j,k \le d}}  \Ll|  \frac{\dr \cGh}{\dr y_j}(y)  \Qx_{jk} \na_k \Gh(y-x)  -  \frac{\dr \cGh}{\dr y_j}(y)\Qx_{jk} \frac{\dr \cGh}{\dr y_k}(y-x) \Rr| \\
 \lesssim \frac{\log |x|_*}{|x|_*^{d-1}}.
$$
Moreover, one can check that
$$
 \sum_{\substack{y \in \Z^d \setminus \{0,x\} \\ 1 \le j,k \le d}}  \Ll| \frac{\dr \cGh}{\dr y_j}(y)\Qx_{jk} \frac{\dr \cGh}{\dr y_k}(y-x) - \int_{y+[0,1]^d}  \frac{\dr \cGh}{\dr y_j}(y')\Qx_{jk} \frac{\dr \cGh}{\dr y_k}(y'-x) \ {\rm d}y' \Rr| \lesssim \frac{\log |x|_*}{|x|_*^{d-1}}.
$$
In these computations, we have been forced to drop some terms indexed by $y \in \{0,x\}$. But it is easy to check that these terms are negligible, for example
$$
 \sum_{\substack{y \in  \{0,x\} \\ 1 \le j,k \le d}} \Ll| \int_{y+[0,1]^d}  \frac{\dr \cGh}{\dr y_j}(u)\Qx_{jk} \frac{\dr \cGh}{\dr y_k}(u-x) \ \d u \Rr| \lesssim \frac{1}{|x|_*^{d-1}} \qquad (x \in \Z^d \setminus \{0\}),
$$
so the proof is complete.
\end{proof}



%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
%
\appendix
\section{Basic estimates on discrete convolutions and Green functions}
\label{s:append}

\begin{prop}
\label{p:sums}
For every $\alpha > d$ and $\beta \in (0,\alpha]$,
$$
\sum_{y \in \Z^d} \frac{1}{|y|_*^\alpha \ |y-x|_*^\beta} \lesssim \frac{1}{|x|_*^\beta},
$$
while for $\beta \in (0,d]$,
$$
\sum_{y \in \Z^d} \frac{1}{|y|_*^d \ |y-x|_*^\beta} \lesssim \frac{\log|x|_*}{|x|_*^\beta}.
$$
(In both statements, the sign $\lesssim$ hides a multiplicative constant that does not depend on $x \in \Z^d$.)
\end{prop}
\begin{proof}
We give a unified proof of these two results, although it will be apparent that the proof of the first statement alone can be slightly simplified. We thus assume $\alpha \ge d$ and $\beta \in (0,\alpha]$. We decompose the sum over $y \in \Z^d$ according to whether $|y| \ge 2|x|$ or not. If $|y| \ge 2|x|$, then $|y-x| \ge |y|/2$, and thus
$$
\sum_{|y| \ge 2|x|} \frac{1}{|y|_*^\alpha \ |y-x|_*^\beta} \lesssim \sum_{|y| \ge 2|x|} \frac{1}{|y|_*^{\alpha+\beta}} \lesssim \frac{1}{|x|_*^{\alpha + \beta - d}} \le \frac{1}{|x|_*^{\beta}}
$$
(here and below, we understand that $y$ is the variable of summation). We split the rest of the sum into two parts along the condition $|y-x| \ge |x|/2$. This gives us two contributions, the first of which is
$$
\sum_{\substack{|y| \le 2 |x| \\ |y-x| \ge |x|/2}} \frac{1}{|y|_*^\alpha \ |y-x|_*^\beta} \lesssim \frac{1}{|x|_*^\beta} \sum_{|y| \le 2 |x|} \frac{1}{|y|_*^\alpha},
$$
This last sum is uniformly bounded if $\alpha > d$, while it is bounded by $\log |x|_*$ if $\alpha = d$. For the second contribution to be considered, note that $|y-x| \le |x|/2$ implies that $|y| \ge |x|/2$, and thus
$$
\sum_{\substack{|y| \le 2 |x| \\ |y-x| \le |x|/2}} \frac{1}{|y|_*^\alpha \ |y-x|_*^\beta} \lesssim \frac{1}{|x|_*^\alpha} \sum_{|y-x| \le |x|/2}  \frac{1}{|y-x|_*^\beta}.
$$
Up to a constant, this last sum is bounded by
$$
\left|
\begin{array}{ll}
1 & \text{ if } \beta > d, \\
\log |x|_* & \text{ if } \beta = d, \\
|x|_*^{d-\beta} & \text{ if } \beta < d.
\end{array}
\right.
$$
Thus, this second contribution is always at most of the order of the first, and this concludes the proof.
\end{proof}
\begin{rem}
\label{r:logs}
The proof of Proposition~\ref{p:sums} can be adapted to yield, for every $\beta \in (0,d]$,
$$
\sum_{y \in \Z^d} \frac{\log|y|_*}{|y|_*^d \ |y-x|_*^\beta} \lesssim \frac{\log^2|x|_*}{|x|_*^\beta}.
$$
\end{rem}


\begin{prop}
For every $k \in \{1,\ldots, d \}$,
\label{p:discrete-to-continuous}
$$
\Ll|\na_k \Gh(x) - \frac{\dr}{\dr x_k} \cGh(x) \Rr| \lesssim \frac{1}{|x|^d}.
$$
\end{prop}
\begin{proof}
Recall that $\Ah$ is a diagonal matrix, the diagonal entries of which we denote by $A_{\mathsf{h},1}, \ldots, A_{\mathsf{h},d}$. For $p \in [-\pi,\pi]^d$, let 
$$
s(p) = 2 \sum_{j = 1}^d A_{\msf{h},j}(1-\cos(p_j)).
$$
Using Fourier transforms, one can represent the Green function $\Gh$ as
$$
\Gh(x) = \frac{1}{(2\pi)^d} \int_{\T} \frac{e^{-i p \cdot x} }{s(p)} \ \d p,
$$
where $\T = [-\pi,\pi]^d$. Similarly,
$$
\na_j \Gh(x) = \frac{1}{(2\pi)^d} \int_{\T} \frac{(e^{-i p_j} - 1)  }{s(p)} \ e^{-i p \cdot x} \ \d p.
$$
Let $\eta(x) = (2\pi)^{-d/2} e^{-|x|^2/2}$. We note that
$$
\Ll|\frac{\dr \cGh}{\dr x_k} (x) - \Ll(\frac{\dr \cGh}{\dr x_k} * \eta\Rr) (x) \Rr| \lesssim \frac{1}{|x|^d},
$$
where $*$ denotes the convolution. This can be seen for instance using the explicit formula for the Green function,
$$
\cGh(x) = \frac{1}{(d-2) \gamma_d \ |\det(\Ah)| \ (x \cdot \Ah^{-1} x)^{(d-2)/2}},
$$
where $\gamma_d$ denotes the area measure of the unit sphere. The regularization by convolution permits us to write down the Fourier representation
$$
\Ll(\frac{\dr \cGh}{\dr x_k} * \eta\Rr) (x) = \frac{1}{(2\pi)^d} \int_{\R^d} - \frac{i p_j  }{p \cdot \Ah p} \ e^{-|p|^2/2}\ e^{-i p \cdot x} \ \d p.
$$
In order to prove the proposition, it thus suffices to show that
$$
\Ll| \int_{\T} \frac{(e^{-i p_j} - 1)  }{s(p)} \ e^{-i p \cdot x} \ \d p - \int_{\R^d} - \frac{i p_j  }{p \cdot \Ah p} \ e^{-|p|^2/2}\ e^{-i p \cdot x} \ \d p \Rr| \lesssim \frac{1}{|x|^d}.
$$
We select a smooth cut-off function $\chi(p)$ that is equal to one near $p=0$ and is compactly supported in $\T$.
We use it to split the left-hand side into
%
$$
\int_{\T}(1-\chi)(p)\frac{(e^{-i p_j} - 1)  }{s(p)} \ e^{-i p \cdot x} \ \d p\quad\mbox{and}\quad
\int_{\mathbb{R}^d}f(p)\ e^{-i p \cdot x} \ \d p,
$$
%
where
$$
f(p) = \chi(p)\frac{(e^{-i p_j} - 1)  }{s(p)} + \frac{i p_j  }{p \cdot \Ah p} \ e^{-|p|^2/2}
$$
%
can be considered to be defined on all $\mathbb{R}^d$. By the properties of $\chi$,
$(1-\chi)(p)\frac{(e^{-i p_j} - 1)  }{s(p)}$ is a smooth periodic function on $\T$, so that we obtain by
integrations by parts that
$$
\int_{\T} (1-\chi)(p)\frac{(e^{-i p_j} - 1)  }{s(p)}\ e^{-i p \cdot x} \ \d p
$$
decays faster than any negative power of $|x|$. Hence it suffices to show that
\begin{equation}
\label{e:toshowgreen}
\Ll| \int_{\mathbb{R}^d} f(p) \ e^{-ip \cdot x} \ \d p \Rr| \lesssim \frac{1}{|x|^d}.
\end{equation}
One can decompose $f$ as
$$
f(p) = -\frac{p_j^2}{2 p \cdot \Ah p} + \td{f}(p),
$$
so that $\td{f}$ is ``more regular'' than $f$ close to the origin. One can then show by integration by parts that
$$
\Ll|  \int_{\mathbb{R}^d} -\frac{p_j^2}{2 p \cdot \Ah p}\ e^{-ip \cdot x} \ \d p - \Ll(\frac{\dr^2 \cGh}{\dr x_j^2}* \eta\Rr) (x)\Rr| \lesssim \frac{1}{|x|^d}
$$
and 
$$
\Ll| \int_{\mathbb{R}^d} \td{f}(p) \ e^{-ip \cdot x} \ \d p \Rr| \lesssim \frac{1}{|x|^d}
$$
for any $x\in\mathbb{R}^d$.
Since $(\dr^2 \cGh/\dr x_j^2 * \eta)(x) \lesssim |x|^{-d}$, the proof is complete.
\end{proof}



\noindent \textbf{Acknowledgements.} We would like to thank Marek Biskup for stimulating discussions about this problem.




\end{document}








